Author’s response to reviews

Title: A randomized cross over trial of tolerability and compliance of a micronutrient supplement with low iron separated from calcium vs high iron combined with calcium in pregnant women

Title: [ISRCTN1221387089824805] Eric Ahn(2), Nicholas Pairaudeau (2), Nicholas Pairaudeau, Jr.(2) Yves Cerat, Bernard Couturier, Andre Fortier, Eric Paradis, Gideon Koren(1) 2Division of Clinical Pharmacology/Toxicology, The Hospital for Sick Children, 1The University of Toronto, 2North York General Hospital, Toronto; and Clinique de Gynecologie-Obstetrique Pierre Boucher, Longueuil, Quebec

Authors:

- Eric Ahn (eric.ahn@utoronto.ca)
- Nicholas Pairaudeau (npairaudeau@hotmail.com)
- Nicholas Pairaudeau Jr. (npairaudeau@hotmail.com)
- Yves Cerat (yvescerat@videotron.ca)
- Bernard Couturier (bcouturier@videotron.ca)
- Andre Fortier (jandrej.fortier@videotron.ca)
- Eric Paradis (ericparadis65@hotmail.ca)
- Gideon Koren (gkoren@sickkids.ca)

Version: 9 Date: 31 March 2006

Author’s response to reviews:

As suggested, we have made changes to the following:

Author list - are in order and full names have been spelled out.
References - All references cited in the text are in square brackets.
Abstract - Have been structured into Background, Methods, Results and Conclusion.